Author's response to reviews

Title: A unique combination of testicular seminomatous mixed germ cell tumor with choriocarcinoma and teratoma with secondary somatic malignancy: a case report

Authors:

Amandeep Aneja (amandeep.aneja@tuhs.temple.edu)
Siddharth Bhattacharyya (Siddharth.Bhattacharyya@tuhs.temple.edu)
Jack Mydlo (Jack.mydlo@tuhs.temple.edu)
Susan Inniss (Susan.inniss@tuhs.temple.edu)

Version: 2 Date: 17 September 2013

Author's response to reviews:

Dear Editors,

Please find below the list of revisions made to the manuscript.
1. The whole text is numbered as a footer.
2. Text “A case report” is added to the title.
3. “L” and “S” changed to lower case in case presentation section of abstract.
4. Ethnicity of the patient is added in abstract and main text.
5. Uniformity of “beta-hCG” is maintained throughout.
6. Keywords have been moved to after the conclusion in abstract.
7. ‘Introduction” of the main text has been moved to the next page.
8. “Man” changed to “male” in case presentation of main text.
9. Figures citation in the text have been added.
10. “echogenicity”, “epididymis” and “teratomatous” have been corrected.
11. Last available follow up of the case has been removed from discussion section and added to the case report section.
12. Lymph node status has been mentioned.
13. “Conclusion” has been added.
14. A list of abbreviations used has been added after the conclusion section and expanded in the text wherever applicable e.g. IHC,PLAP,EMA
15. References have been revised, including all named authors up to first 30 before adding et al.
16. Questions regarding Bleomycin therapy have been addressed to.
17. Definition of teratoma with somatic type malignancy included in the discussion.
18. We do agree with the possibility raised by Reviewer #1 Dr. Adam Zarkadoulias in point no. 5 that the metastatic component could be choriocarcinoma but unfortunately, biopsies from the suspected metastatic sites were not diagnostic.